Author's response to reviews

Title: The contribution of qualitative research in designing a complex intervention for secondary prevention of coronary heart disease in two different healthcare systems [ISRCTN24081411]

Authors:
Mairead Corrigan (m.corrigan@qub.ac.uk)
Margaret E Cupples (m.cupples@qub.ac.uk)
Susan M Smith (susmith@tcd.ie)
Molly Byrne (molly.byrne@nuigalway.ie)
Claire S Leatham (c.leathem@qub.ac.uk)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Andrew W Murphy (andrew.murphy@nuigalway.ie)

Version: 5 Date: 12 July 2006

Author's response to reviews:

Dear Editor,

I accidentally resubmitted an old version of the manuscript this morning so I would ask you to disregard this earlier submission as I am resubmitting the final version with requested revisions with this cover letter. I have made the following changes...

I have changed some of the authors’ addresses as these were inaccurate. I transferred the Trial Registry number from the title to the abstract. In the main manuscript I replaced the Introduction for Background. I removed the vertical lines in the tables and in Figure 1. I also changed the footnote in Figure 1 so that it is no wider than the image and cropped the image accordingly. I hope that these changes are acceptable and please contact me if you have any further queries.

Yours sincerely

Mairead Corrigan